Author’s response to reviews

Title: Outcomes of Orbital Malignancies Treated with Eye-Sparing Surgery and Adjuvant Particle Radiotherapy: a retrospective study

Authors:
Weixu Hu (weixu.hu@sphic.org.cn)
Jiyi Hu (jiyi.hu@sphic.org.cn)
Jing Gao (jing.gao@sphic.org.cn)
Jing Yang (jing.yang@sphic.org.cn)
Xianxin Qiu (xianxin.qiu@sphic.org.cn)
Lin Kong (lin.kong@sphic.org.cn)
Jiade Lu (jiade.lu@sphic.org.cn)

Version: 2 Date: 19 Jul 2019

Author’s response to reviews:

Dear Dr. Jones,

Thank you very much for assessing our manuscript titled “Outcomes of Orbital Malignancies Treated with Eye-Sparing Surgery and Adjuvant Particle Radiotherapy: a retrospective study”. We appreciate your time and efforts. Based on the comments, we have revised the manuscript and have addressed all the comments. The point-by-point responses to the editor’s comments are presented below.

If you have any questions about our work, please do not hesitate to contact us. We are looking forward to hearing from you soon.

Best wishes,

Drs. Lin Kong and Jiade J. Lu

Shanghai Proton and Heavy Ion Center (SPHIC)
4365 Kangxin Road, Pudong, Shanghai 201321
Web: www.sphic.org.cn   Tel: +86.21.38296505
Replies to editor comments 1:

Please insert the heading 'Declarations' for the Declarations section, above the 'Ethics approval and consent to participate' section of your manuscript.

Response: Thank you for your format change comments, we have added the title ‘Declarations’ for the Declarations section.

Replies to editor comments 2:

In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We have stated “the funders had no role in the design of the study, in the data collection, analysis, or interpretation of data, in the decision to publish, or in the writing of the manuscript” in Funding section.

Replies to editor comments 3:

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: In the Authors’ Contributions section, full initials of authors' names have been used instead of the full names.

Replies to editor comments 4:

Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Response: We have added the title ‘Figure Legends’ and move the Figure titles/legends to the end of the manuscript after the References.

Replies to editor comments 5:

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional SuppleF5pmentary files are cited within the text.
Response: Since there are no extra figures or tables that need to be upload, we have reuploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.